Author’s response to reviews

Title: The natural history of, and risk factors for, progressive Chronic Kidney Disease (CKD): the Renal Impairment In Secondary care (RIISC) study; rationale and protocol

Authors:

Stephanie Stringer (stephanie.stringer@uhb.nhs.uk)
Praveen Sharma (praveen.sharma@nhs.net)
Mary Dutton (mary.dutton@uhb.nhs.uk)
Mark Jesky (mark.jesky@uhb.nhs.uk)
Khai Ng (KhaiPing.Ng@uhb.nhs.uk)
Okdeep Kaur (okdeep.kaur@uhb.nhs.uk)
Iain Chapple (I.L.C.CHAPPLE@bham.ac.uk)
Thomas Dietrich (t.dietrich@bham.ac.uk)
Charles Ferro (charles.ferro@uhb.nhs.uk)
Paul Cockwell (paul.cockwell@uhb.nhs.uk)

Version: 3 Date: 2 February 2013

Author’s response to reviews: see over
Dear Editor,

Re: Manuscript reference 4388015998726383 amendments

Please find attached a manuscript entitled “The natural history of, and risk factors for progressive Chronic Kidney Disease (CKD): the Renal Impairment In Secondary care (RIISC) study; rationale and protocol”.

Please find attached the revised manuscript, the changes made are as follows;

1. We have inserted the statement that informed consent was obtained from participants in line with GCP guidance
2. We have inserted a statement acknowledging the contribution of our funders in the acknowledgments section.

Yours faithfully

Stephanie Stringer (on behalf of all the authors)